Author's response to reviews

Title: Functional characterisation of the TSC1-TSC2 complex to assess multiple TSC2 variants identified in single families affected by tuberous sclerosis complex

Authors:

Mark Nellist (m.nellist@erasusmc.nl)
Ozgur Sancak (Ozgur.Sancak@dsm.com)
Miriam Goedbloed (m.goedbloed@erasusmc.nl)
Alwin Adriaans (a.adriaans@erasusmc.nl)
Marja Wessels (m.w.wessels@erasusmc.nl)
Anneke Maat-Kievit (j.a.maat@erasusmc.nl)
Marieke Baars (m.j.baars@amc.uva.nl)
Charlotte Dommering (cj.dommering@vumc.nl)
Ans van den Ouweland (a.vandenouweland@erasusmc.nl)
Dicky Halley (d.halley@erasusmc.nl)

Version: 4 Date: 1 February 2008

Author's response to reviews: see over
The Editor,
BMC Medical Genetics,

4th February 2008

re: revision manuscript 2200778961635001

Dear Dr. Manning,

Many thanks for sending the comments of the two reviewers on our revised manuscript 2200778961635001, 'Functional characterisation of the TSC1-TSC2 complex to assess multiple TSC2 variants identified in single families'. We have now included the extra figure (Figure 2B), as requested, and updated the manuscript text and figure legends accordingly.

We hope that the revised manuscript will now be acceptable for publication.

Yours faithfully,

Mark Nellist (on behalf of all co-authors)